Author's response to reviews

Title: A novel sero-genetic approach determines the community prevalence of celiac disease and informs improved diagnostic pathways

Authors:

Robert Anderson (bob@immusant.com)
Margaret J Henry (margh@barwonhealth.org.au)
Roberta Taylor (bobbie3@optusnet.com.au)
Emma L Duncan (emma.duncan@uq.edu.au)
Patrick Danoy (patrick.danoy@roche.com)
Marylia J Costa (costamarylia@yahoo.com.au)
Kathryn Addison (Kathryn.addison1@gmail.com)
Jason A Tye-Din (tyedin@wehi.edu.au)
Mark A Kotowicz (markk@barwonhealth.org.au)
Ross E Knight (reknight@geelonggastro.com.au)
Wendy Pollock (Wendy.Pollock@healthscope.com.au)
Geoffrey C Nicholson (geoff.nicholson@uq.edu.au)
Ban-Hock Toh (Ban-Hock.Toh@monash.edu)
Matthew A Brown (matt.brown@uq.edu.au)
Julie A Pasco (juliep@BarwonHealth.org.au)

Version: 2 Date: 12 July 2013

Author's response to reviews:

See letter to the Editor